Author’s response to reviews

Title: Factors influencing turnover intention among primary care doctors: a cross-sectional study in Chongqing, China

Authors:

Tong Wen (wentong16@yeah.net)
Yan Zhang (cqmudzy@163.com)
Xue Wang (m13320256378_1@163.com)
Guo Tang (t295920124@icloud.com)

Version: 3 Date: 31 Jan 2018

Author’s response to reviews:

Dear Dr. Dieleman,

We thank you and the two reviewers for these comments on our manuscript entitled “Factors influencing turnover intention among primary care doctors: a cross-sectional study in Chongqing, China”. We had made a careful editing of this manuscript according to the reviewers’ comments. The language of this manuscript also had been edited again by native English speakers. We hope that this version of the manuscript could be published on Human Resources for Health. We have made a point-by-point response to the reviewers’ comments below.

Reviewer 1#

Reviewer’s comment

This version of the manuscript is much improved than the last one that I had reviewed. The language is clearer and the discussion section is much stronger. I commend the authors for making these extensive changes and adjustments to the manuscript.

Author’s response

Thank you for your review and comments.

Reviewer 2#

Reviewer’s comment
The paper is informative and well laid out. The paper reads as if has been translated and some words are not correct. The paper requires more than copy editing. The content must also be edited before it is suitable for publication.

Author’s response

We appreciate the reviewer’s comments. We correct some misused words and made a careful editing for this manuscript. Please see the paper.

Finally, we hope that these responses are satisfactory and thank you again for your consideration.

Best wishes.

Yours sincerely,

Yan Zhang
E-mail: cqmudzy@163.com